Author’s response to reviews

Title: A comparison of course-related stressors in undergraduate problem-based learning (PBL) versus non-PBL medical programmes

Authors:

  Alexander D Lewis (ADL548@bham.ac.uk)
  Darryl A Braganza Menezes (DAB531@bham.ac.uk)
  Helen E McDermott (HEM507@bham.ac.uk)
  Louise J Hibbert (LJH555@bham.ac.uk)
  Sarah-Louise Brennan (SLB578@bham.ac.uk)
  Elizabeth E Ross (EER573@bham.ac.uk)
  Lisa A Jones (l.a.jones@bham.ac.uk)

Version: 5 Date: 9 September 2009

Author’s response to reviews: see over
9th September 2009

Editorial Production Team
BMC Medical Education

Dear Editorial Production Team

Thank you for your email of 8th September 2009. We are delighted that BMC Medical Education will publish our manuscript *A comparison of course-related stressors in undergraduate problem-based learning (PBL) versus non-PBL medical programmes* (MS 1219950777251280). We confirm that we have made the following changes to the format of our manuscript as requested:

**Major Revisions**

1. The list of authors in the manuscript is now written exactly as they are in the submission system.
2. All vertical elements and merged cells have been removed from the tables.
3. The references are cited in the text using consecutive numbers in square brackets without use of superscript.

**Minor Revisions**

1. The table placement text has been removed from the manuscript.

Yours Sincerely

Alexander D Lewis
Darryl A Braganza Menezes
Helen E McDermott
Louise J Hibbert
Sarah-Louise Brennan
Elizabeth E Ross
Lisa A Jones